Author's response to reviews

Title: Natural history of colonization with methicillin-resistant Staphylococcus aureus (MRSA) and vancomycin-resistant Enterococcus (VRE): a systematic review

Authors:

   Erica S Shenoy (eshenoy@partners.org)
   Molly L Paras (mparas@partners.org)
   Farzad Noubary (farzad.noubary@gmail.com)
   Rochelle P Walensky (rwalensky@partners.org)
   David C Hooper (dhooper@partners.org)

Version: 2
Date: 4 March 2014

Author's response to reviews: see over
March 4, 2014

Philippa Harris
Executive Editor
BMC Infectious Diseases

Re: Manuscript 6613325711010633

Dear Dr. Harris,

We appreciate the opportunity to submit our revised manuscript “Natural history of colonization with methicillin-resistant Staphylococcus aureus (MRSA) and vancomycin-resistant Enterococcus (VRE): a systematic review” for consideration for publication in BMC Infectious Diseases.

In the accompanying Response to Reviewers, we have addressed all comments by the editor and referees in a point-by-point format, indicating the page and line number of the modifications in the revised manuscript.

All authors have approved this submission and have no relevant conflicts of interest to report.

Thank you for your consideration and for the extension provided to make the necessary changes.

Sincerely,

Erica S. Shenoy, MD, PhD
Assistant Chief, Infection Control Unit
Massachusetts General Hospital
Instructor in Medicine
Harvard Medical School
eshenoy@partners.org

David C. Hooper, MD
Associate Chief, Div. Infectious Diseases
Chief, Infection Control Unit
Massachusetts General Hospital
Professor of Medicine
Harvard Medical School
dhooper@partners.org